Author's response to reviews

Title: A Study of a Culturally Focused Psychiatric Consultation Service for Asian American and Latino American Primary Care Patients with Depression

Authors:

Nhi-Ha T Trinh (ntrinh@partners.org)
Carlos A Bedoya (abedoya@partners.org)
Trina E Chang (techang@partners.org)
Katherine Flaherty (katherineflaherty17@gmail.com)
Maurizio Fava (mfava@partners.org)
Albert Yeung (ayeung@partners.org)

Version: 4 Date: 30 September 2011

Author's response to reviews: see over
September 26, 2011

BMC Series Editorial Production Team

To whom it may concern:

Thank you for your helpful comments. Please see below for the details of changes made:

**Major revisions (we require the author to make these changes)**

**References: please provide a functioning link to reference 27.**

A new link has been provided for this reference.

**Affiliations: please link Trina E. Chang to an affiliation.**

Sorry for this omission; it has been corrected.

**Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)**

**Author list: please remove any author titles and qualifications.**

Please clarify—are you referring to the footnoted item on the cover page, “these two authors contributed equally to the work”? If so, we would prefer to keep this, as Drs. Trinh and Bedoya were co-first authors on this grant proposal.

**Competing interests: please ensure that the font type and size are consistent through this section.**

This has been amended.

**Table: please remove the visible vertical lines from your table.**

We have removed the vertical lines from the table.
Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures.

We have tried to modify the figure accordingly. Please advise.

Additional Files: please remove these.

We have removed prior files. Are there additional files to which you are referring? Please advise and clarify if there are additional files to be deleted.

Please let us know if there is additional information you may require. Thanks for your consideration.

Sincerely,

Nhi-Ha Trinh MD MPH